

Author's response to reviews

Title: Trunk rotational strength asymmetry in adolescents with idiopathic scoliosis: an observational study

Authors:

Kevin L McIntire (kmcintire@kumc.edu)
Marc A Asher (masher@kumc.edu)
Douglas C Burton (dburton@kumc.edu)
Wen Liu (wliu@kumc.edu)

Version: 4 **Date:** 6 July 2007

Author's response to reviews:

Removed the redundant 'figure' in the figure legends.